

# GeoHealth

# **RESEARCH ARTICLE**

10.1029/2024GH001092

#### **Key Points:**

- No general patterns found describing the heatwave characteristics-mortality association
- The heatwave characteristics-mortality coupling needs a local scale analysis due to its complex and strong dependence on local properties
- The health indices recovery factor and excess heat factor do not always represent the heatwave-mortality association

**Supporting Information:** 

Supporting Information may be found in the online version of this article.

#### **Correspondence to:**

L. Paredes-Fortuny, laurapf@ceam.es

#### Citation:

Paredes-Fortuny, L., Salvador, C., Vicedo-Cabrera, A. M., & Khodayar, S. (2024). Geographical patterns in mortality impacts due to heatwaves of different characteristics in Spanish cities. *GeoHealth*, 8, e2024GH001092. https:// doi.org/10.1029/2024GH001092

Received 3 MAY 2024 Accepted 6 JUL 2024

#### **Author Contributions:**

Conceptualization: Ana M. Vicedo-Cabrera, Samira Khodayar Data curation: Laura Paredes-Fortuny Formal analysis: Laura Paredes-Fortuny Funding acquisition: Samira Khodayar Investigation: Laura Paredes-Fortuny, Coral Salvador, Ana M. Vicedo-Cabrera, Samira Khodayar Methodology: Coral Salvador, Ana M. Vicedo-Cabrera, Samira Khodayar Software: Laura Paredes-Fortuny Supervision: Ana M. Vicedo-Cabrera, Samira Khodayar Visualization: Laura Paredes-Fortuny

© 2024. The Author(s). GeoHealth published by Wiley Periodicals LLC on behalf of American Geophysical Union. This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# **Geographical Patterns in Mortality Impacts Due To Heatwaves of Different Characteristics in Spanish Cities**

Laura Paredes-Fortuny<sup>1</sup>, Coral Salvador<sup>2,3,4</sup>, Ana M. Vicedo-Cabrera<sup>3,4</sup>, and Samira Khodayar<sup>1</sup>

<sup>1</sup>Mediterranean Center for Environmental Studies (CEAM), Valencia, Spain, <sup>2</sup>Centro de Investigación Mariña, Environmental Physics Laboratory (EPhysLab), Universidade de Vigo, Ourense, Spain, <sup>3</sup>Institute of Social and Preventive Medicine (ISPM), University of Bern, Bern, Switzerland, <sup>4</sup>Oeschger Centre for Climate Change Research, University of Bern, Bern, Switzerland

**Abstract** The impact of heatwaves (HWs) on human health is a topic of growing interest due to the global magnification of these phenomena and their substantial socio-economic impacts. As for other countries of Southern Europe, Spain is a region highly affected by heat and its increase under climate change. This is observed in the mean values and the increasing incidence of extreme weather events and associated mortality. Despite the vast knowledge on this topic, it remains unclear whether specific types and characteristics of HW are particularly harmful to the population and whether this shows a regional interdependency. The present study provides a comprehensive analysis of the relationship between HW characteristics and mortality in 12 Spanish cities. We used separated time series analysis in each city applying a quasi-Poisson regression model and distributed lag linear and non-linear models. Results show an increase in the mortality risk under HW conditions in the cities with a lower HW frequency. However, this increase exhibits remarkable differences across the cities under study not showing any general pattern in the HW characteristics of each city pointing out the crucial need to examine and understand on a local scale the HW characteristics and the HW-mortality relationship for an efficient design and implementation of prevention measures.

**Plain Language Summary** Heatwaves (HWs) are episodes of extreme heat sustained in time with devastating socio-economic impacts. Due to their global magnification, the interest in their impacts on human health has increased. Spain, in Southern Europe, is a climate change hot spot, particularly in relation to increasing temperature extremes. Despite the relevance of the topic, it is still unclear if there are particular characteristics of heatwaves with a larger impact on mortality. In the present study, we analyze the relationship between heatwaves' characteristics and mortality risk in 12 Spanish cities. Results show no general pattern for the relationship between mortality and heatwave characteristics over the 12 cities under study, but local relations point out the need for local studies to accurately assess the relationship between heatwave characteristics and mortality for an efficient implementation of prevention measures.

# 1. Introduction

Heatwaves (HW) are extreme events of high temperature that pose a threat to human health (Barriopedro et al., 2023; Khodayar & Paredes-Fortuny, 2024; Paredes-Fortuny & Khodayar, 2023). Spain, located in the western region of the Mediterranean Sea, is particularly affected by this phenomenon. The Mediterranean is a climate change hot spot warming 20% faster than other areas of the globe (Lionello & Scarascia, 2018; MedECC, 2020). The IPCC reports indicate that HWs will become more frequent and severe in the Mediterranean in the coming years (IPCC et al., 2021). A joint spatiotemporal investigation of the HWs in Spain, reveals a magnification of this phenomenon resulting from a growing population of the most extreme HWs (Paredes-Fortuny & Khodayar, 2023). Additionally, that analysis suggests that the most damaging HWs are not necessarily the most intense but can be extremely long-lasting or have a low recovery factor (Pereira et al., 2017, RF) stressing the importance of considering the overall HW characteristics when studying the association between HWs and mortality. That study also shows that there are relevant local differences in the HW characteristics. On the coast, HWs are of higher intensity but occur at a lower frequency. These local particularities must be considered when the impacts on the population are studied. In general, higher intensity and duration HWs' are expected in Spain in the future (Lorenzo et al., 2021; Viceto et al., 2019).





10.1029/2024GH001092

Writing – original draft: Laura Paredes-Fortuny Writing – review & editing: Coral Salvador, Ana M. Vicedo-Cabrera, Samira Khodayar Parallel to this, there is substantial evidence showing an increase in mortality associated with HWs in many areas of the globe (Ballester et al., 2023; Kollanus et al., 2021; Lo et al., 2022; Lopez-Bueno et al., 2020; Moraes et al., 2022; Ruiz-Páez et al., 2023; Ruuhela et al., 2020; Silveira et al., 2023; Yang et al., 2019). The European HW of 2003 was a record-breaking case, which was responsible for 70,000 additional deaths in Europe (Robine et al., 2008). Recent epidemiological investigations have shown that HWs, and more generally heat, are particularly impactful in the Mediterranean region (Zhao et al., 2021). Despite the recent efforts to protect the population from extreme heat, in the past summer of 2022 about 5,000 people died in Spain due to extreme heat (Tobías et al., 2023). In Madrid and Barcelona, the two most populated cities of Spain, heat has been responsible for about 16% and 22% of the hospitalisations resulting in death due to respiratory disease during the 2006–2019 period (Achebak et al., 2023).

During the last years, mortality associated with HWs in the Mediterranean area and Spain have shown a decrease in some cities (Díaz et al., 2018; Linares et al., 2015; Mirón et al., 2015) but an increase in others (Díaz et al., 2018). Several factors have been proposed, namely the introduction of heat-health prevention plans (Díaz et al., 2018), the income level which is strongly related to the existence of air conditioning units (Linares et al., 2015; Lopez-Bueno et al., 2020; Mirón et al., 2015), and the proportion of older adults which is reported as a risk factor (Arsad et al., 2022; Kollanus et al., 2021; Lopez-Bueno et al., 2020; McKenna et al., 2023). The large spatial heterogeneity of the heat-mortality association (Arsad et al., 2022; D'Ippoliti et al., 2010; Kang et al., 2020; Ruuhela et al., 2020) highlights the importance of understanding which factors are the most relevant to this association and how they are related to the mortality increase for the effective development of adaptation and protection plans.

Several previous investigations already examined the effect of heat and HWs on mortality in Spain (Carmona et al., 2017; Díaz et al., 2018; Roldán et al., 2016; Royé et al., 2020). However, current knowledge does not consider the changes in the mortality risk associated with different combinations of the HW duration with the HW intensity, minimum temperature, excess heat factor (EHF, Nairn & Fawcett, 2014), and RF (Díaz et al., 2018; Linares et al., 2015; Miron et al., 2015). Moreover, the later indices, EHF (Excess Heat Factor; Nairn & Fawcett, 2014, EHF) and RF (Recovery Factor; Pereira et al., 2017, RF) are widely used in meteorology, but a quantitative evaluation of their adequacy to describe the increase in mortality in different climatic, demographic, and socioeconomic conditions is lacking. The EHF measures the HW severity taking into account the human climatization capability by considering the recent past temperatures as a reference. The RF quantifies the thermal comfort considering nighttime temperatures which are related with the capacity of the human body to recover from the exposure to daytime high temperatures. To the authors' knowledge, this is the first analysis addressing these two aspects using a set of HWs identified with a common definition from the Spanish MetService, AEMET. The present analysis at the city scale includes 12 province capitals covering several decades, different climates, and population densities. This evidence would help improve preparedness for future HW events by adapting current mitigation and prevention measures to the singularities of the local-scale HW impacts (Linares et al., 2015; López-Bueno et al., 2020).

#### 2. Data and Methodology

#### 2.1. Area of Study and Data Collection

The study was performed independently in 12 Spanish province capitals (Figure 1a). The ensemble of cities illustrates a diversity of climates and conditions in Spain, with coastal and inland, Mediterranean and Atlantic, lower and higher population density cities (see Table S1 in Supporting Information S1 for more details of the cities' climatology). Time series of daily all-cause mortality data of the 12 studied cities was collected from Instituto Nacional de Estadística (INE, Spanish Statistics Institute). Mortality data consists of the reported daily mortality due to all causes in each city for the extended summers (May–September) from 1975 to 2019. A description of the mortality data, in terms of total mortality and mortality rate (per 105 inhabitants) is provided in Table S1 and Figure S1 of the Supporting Information S1.

Daily maximum and minimum temperatures (Tmax and Tmin, respectively) were retrieved from the ROCIO\_IBEB data set, a 5 km gridded observational data set covering peninsular Spain and the Balearic Islands from Agencia Estatal de Meteorología (AEMET, Spanish MetService). This data set includes information on 2-m Tmax and Tmin obtained from the interpolation of the data derived by 1,800 thermometric stations covering Spain and the Balearic Islands with the climatology. The climatology used for the interpolation is based on the







**Figure 1.** (a) Cities under study grouped by climatological and geographical characteristics: Mediterranean coast (orange), Inland mild Tmin (green), Cantabrian Coast (blue), and Inland low Tmin (purple). (b) For each city (*x*-axis), daily minimum (blue), maximum (red) and mean (black) temperature (Tmin, Tmax, and Tmean respectively). Background shadow indicates the aforementioned groups to which each city belongs.

historical analysis of the numerical prediction model HIRLAM. Further details regarding this data set can be consulted in Amblar-Francés et al. (2020). Daily mean temperature (Tmean) is approximated as the mean of Tmax and Tmin. The temporal range of the data used in the present study covers the extended summers (May–September) from 1975 to 2019. Considering the summer mean daily minimum temperature and the geographical location (inland/coast) of each city, four groups are identified namely the Mediterranean coast, the Inland mild daily minimum temperatures, the Cantabrian coast and the Inland low daily minimum temperatures (Figure 1b).

#### 2.2. Heatwave Identification and Characterization

In the present study, HWs were defined following the Spanish Meteorological Service, AEMET, definition (AEMET. Olas de calor en España des de, 2023) as an event in which for at least 3 consecutive days Tmax exceeds the 95th percentile (P95) and affects at least 5% of the area under study. The 5% spatial threshold is used following the approach of Paredes-Fortuny and Khodayar (2023), a comprehensive analysis of the HW phenomena over Spain. Following the HW identification methodology of Paredes-Fortuny and Khodayar (2023), HWs were computed at each grid point of Peninsular Spain and the Balearic Islands (PSBI) for each day of the period under study, extended summers (May–September) from 1975 to 2019. P95 was computed at each gridpoint of PSBI using the Tmax time series of the climatological reference period, May–September 1961–1990, following the WMO recommendations (World Meteorological Organization, 2017). The days that make up a HW are named HWD. Thus, a daily time series of HWD occurrences was obtained for each grid point. Afterward, the HWD time series of the grid points contained in each city were aggregated into a unique time series for each city.



Definitions of the Variables Used in This Study and Their Corresponding Units

Variable name	Definition	Units
HW	Three or more consecutive days with Tmax $>$ P95 affecting 5% of PSBI	Event
HWD	Day t that conforms a HW	Days
HWD Intensity <sup>a</sup>	$\mathrm{Tmax}(t) - P95$	°C
HWD Cumulative duration	Days after the HW onset plus one	Days
HWD Tmin <sup>a</sup>	Tmin(t)	°C
HWD Excess heat factor (EHF, Nairn & Fawcett, 2014) <sup>a</sup>	The EHF is a measure of the HWD severity regarding the magnitude of the intensity and how sudden the temperature rise is $EHI_{sig}(t) = \frac{T(t)+T(t+1)+T(t+2)}{3} - P95$ $EHI_{acc}(t) = \frac{T(t)+T(t+1)+T(t+2)}{3} - \frac{T(t-1)+\dots+T(t-30)}{30}$ $EHF(t) = EHI_{sig}(t) \cdot max(1, EHI_{acc}(t))$	°C <sup>2</sup>
HWD Recovery factor (Karl & Knight, 1997; Pereira et al., 2017) <sup>a</sup>	Measure of human thermal comfort. $RF(t) = Tmax(t) - Tmin(t)$	°C
HWD level	Classification of the HWD indicating the magnitude of the HWD's temperature variables (i.e., intensity, Tmin, EHF, and RF) in combination with the magnitude of the cumulative duration	Unitless

<sup>a</sup>This is the definition for each gridpoint. The corresponding HWD' value is the mean of the city gridpoints affected by the HW on this day.

The obtained HWDs of each city were then characterized independently by their intensity, cumulative duration, and daily minimum temperature (Tmin). Additionally, two health indices namely the excess heat factor (EHF, Nairn & Fawcett, 2014) and the recovery factor (RF, Pereira et al., 2017) were calculated. Table 1 shows the definitions of the above-mentioned HW characteristics and health indices. On a later step and independently for each city, HWDs were classified into levels that indicate the value of each of the HWD's temperature-related variables (i.e., intensity, Tmin, EHF, and RF) in combination with the value of the cumulative duration as follows. Each HWD is given one label for each HWD variable namely intensity, Tmin, RF, EHF, and cumulative duration that can be >P50 or <P50 indicating if the value of the variable is above (>P50) or below (<P50) the 50th percentile of the given city. The threshold was set at 50% to ensure both groups had the same number of HWDs. Afterward, independently for each temperature-related variable (i.e., intensity, Tmin, EHF, and RF), four possible levels were obtained resulting from the combination of the two possible labels of the magnitude of the temperature-related variable and the two possible labels of the cumulative duration. In parallel, HWs were classified into 16 typologies regarding the simultaneous values of the HW' Tmax, Tmin, intensity and duration (the number of days that the HW lasts over a city). These typologies are the combination of high/low values of each of these four HW variables. The threshold used to classify HWs as high or low regarding each variable is the 50th percentile of the HWs of all cities. Thereafter, the number of occurrences and the proportion of each HW type in each city was computed. The detailed description of each HW typology and its frequency in each city is shown in Figure S2 of the Supporting Information S1.

#### 2.3. Assessment of the Heat-Mortality Association

To estimate the heat-mortality association, we applied state-of-the-art epidemiological methods consisting of separate time-series analyses in each city. The analysis is restricted to the extended summer months (i.e., from May to September). Daily mortality was modeled as a discrete random variable that follows a quasi-Poisson distribution. That is a discrete random event with a large sample space whose estimated value is larger than its variance. Three models have been used to evaluate the HW-mortality association with increasing complexity. The mathematical definitions of the models and their terms are shown in the detailed description of the models section of the Text S1 in Supporting Information S1. All of them include a temporal term ( $X_t(t)$ ) which controls seasonal and long-term patterns in the outcome with a natural spline (ns) of time with one degree of freedom per year, a natural spline (ns) of the day of the year (doy) with four degrees of freedom, and an indicator of the day of the week (dow). The heat-mortality association was modeled using distributed lag linear and non-linear models (DLM, DLNM) (Gasparrini et al., 2010) to account for potential lagged effects up to 3 days. In the first model,



Figure 2. For each city, the sum of the number of heat waves (HW) and heat wave days (HWD) identified in the period from May to September 1975 to 2019.

named the overall HW effect  $(Y_{HW})$ , HW's occurrence was modeled with a binary variable,  $X_{HW}(t)$ , thus the  $\beta$  coefficient corresponds to the mortality risk associated with HW occurrence. In the second model, named the additive HW effect model  $(Y_{Tmean+HW})$ , Tmean was introduced with the term  $X_{Tmean}(t)$ , modeled with a DLNM consisting of a natural spline with two knots placed at the 50th and 90th percentiles of the city's Tmean and a lag of 3 days modeled with strata function (i.e., 0 and 1–3 lags). Thus, in this case, the  $\beta$  coefficient of the  $X_{HW}(t)$  represents the additive effect associated with HW event (i.e., more than 2 days of high temperatures) on top of the effect of Tmean (i.e., high temperatures). The choice of 3 days is widely used in literature since it has been found that after 3 days the effect of heat on mortality decays (Guo et al., 2017; Huang et al., 2014). The third model, named the HW variable model ( $Y_{HWvar}$ ), is analogous to the second model replacing the term  $X_{HW}(t)$  with a categorical variable indicating the HWD level,  $X_{level}^i(t)$ . The HW variable model was applied independently four times, one for each of the following HWD variables intensity, Tmin, RF, and EHF. For each of these four HWD variables, there are four possible values of the term  $X_{level}(t)$  that are the combination of the two possible labels of the cumulative duration (<P50/>P50) with the two possible labels of the given HWD variable *i* (<P50/>P50) (where i = intensity, Tmin, RF, and EHF).

The modeling has been done with the R package *dlnm* (Gasparrini, 2011). The association between mortality and the different model components has been reported as relative risk (RR), with the days without HW (non-HWD) as a reference. The RRs associated with each of the model terms HW' occurrence  $(X_{HW}(t))$  and HWD' level  $(X_{level}^i(t))$  was computed as  $RR_k = e^{\beta_k}$ , where k = HW, level *i* and  $\beta_k$  is a numerical value obtained from the *glm* model. Akaike Information Criterion (AIC) tests have been done changing the degrees of freedom of the natural splines.

# 3. Results

#### 3.1. HWs Over Spanish Cities

During the extended summers (May-September) between 1975 and 2019, from 22 to 91 HWs were identified in the 12 cities under study. The lowest HW/HWD frequencies (number of HWs/HWDs per year), up to 59 HWs/310 HWDs, were found in the coastal cities of Bilbao, A Coruña, Valencia, Murcia, and Barcelona. Except for Murcia, these cities presented the largest mortality rates, ranging from about 15,000 to 17,500 deaths/10<sup>5</sup> inhabitants in A Coruña and Bilbao, respectively (Figure 2). The analysis of the HW characteristics occurring in each of the 4 climatological groups (Figures S2 and S3 in Supporting Information S1) namely the Mediterranean coast (Barcelona, Málaga, Murcia, Palma, and Valencia), the Cantabric coast (A Coruña and Bilbao), Inland mild daily minimum temperature (Tmin) (Badajoz, Madrid, Sevilla, and Zaragoza) and Inland low Tmin (Valladolid), reveals that the HWs affecting each climatological group were similar. Generally, HWs belonging to the upper 50th percentile of Tmax occur in the Inland mild group and in Murcia (Mediterranean coast group). In contrast, HWs in the Mediterranean and in the Cantabric coasts and in Valladolid belong to the lowest 50th percentile of Tmax. Interestingly, the HWs affecting the Mediterranean coast group are also of large Tmin (the upper 50th percentile of Tmin) while in Valladolid and in the Cantabric coast group are of low Tmin. The largest durations (the number of days that last a HW) were found in Málaga and Palma. Large differences (up to 10°C) were found in the median values of the RF during HWDs, while the EHF presents a median value between 3 and  $5^{\circ}C^{2}$ , except for Bilbao where  $13^{\circ}C^{2}$ were observed during HWDs.





**Figure 3.** Relative risk (RR) associated with the HW occurrence derived from the overall HW effect model ( $Y_{HW}$ , red points) and the additive HW effect model ( $Y_{Tmean+HW}$ , green points). Error bars indicate 95% confidence intervals. Mortality reference is non-HWDs. The horizontal black line at RR = 1 indicates the threshold above which there is larger mortality associated with HWDs compared with non-HWDs.

#### 3.2. Effect of HWs on Mortality

#### 3.2.1. Association Between HW Occurrence and Mortality

According to the overall HW effect model ( $Y_{HW}$ ), HW's occurrence was associated with an increase in mortality risk (RR > 1) in all the cities (Figure 3, red points). In this model, the daily mean temperature was not included. This increase in the mortality risk presents large differences between cities even within the same climatological group (Figure 1). During HWDs, mortality risk increased from 4.2% [95% CI 0.5, 7.9] in Málaga up to 19.0% [95% CI 10.7, 28.1] in Badajoz (Figure 3, red points), compared to non-HWDs. Confidence intervals were larger in less populated cities such as Badajoz, A Coruña, Valladolid, and Bilbao.

# 3.2.2. Additive Effect of HW Occurrence on Temperature (Tmean + HW)

The independent or additive effect of the HW occurrence on top of Tmean is the effect associated with the HW occurrence when Tmean is included in the model (additive HW effect model,  $Y_{\text{Tmean+HW}}$ ). The mortality risk associated with the additive effect of the HW occurrence increased (RR > 1) in the cities with the 6 lowest HW' frequency, namely Valencia, Bilbao, Murcia, Barcelona, A Coruña and Badajoz. This risk ranged from 4.2% [95% CI 1.7, 6.7] in Barcelona up to 8.0% [3.3, 12.8] in Valencia (Figure 3, green points). These cities with RR > 1 are coastal except for Badajoz. However, the RR decreased in all cities compared to the overall HW effect model (Figure 3, red points).

# 3.2.3. Additive Effect of HW Characteristics

In the following step, the influence of the intensity/Tmin/RF/EHF combined with the cumulative duration was assessed with the HW variable model ( $Y_{HWvar}$ ). Four possible levels result from the combination of the two possible labels of the cumulative duration (above/below the 50th percentile) and the two possible labels of the other given HWD variable (above/below the 50th percentile). Figure 4 shows the RR obtained when considering the cumulative duration together with the other four HWD variables (i.e., intensity, Tmin, RF, and EHF) resulting in different patterns for each city.

Generally, an increase in the mortality risk associated with some levels was observed in the cities that already showed an increase in the mortality risk associated with the additive effect of the HW occurrence on top of the effect of temperature (additive HW effect model). These correspond to cities with low HW frequency (A Coruña, Badajoz, Barcelona, Bilbao, Murcia, and Valencia).

Overall, in some cases detailed in Table S2 of the Supporting Information S1, the most dominant characteristic in the increase/decrease of the mortality risk seemed to be the cumulative duration, while in some others it was the intensity/Tmin/EHF/RF or a specific combination of high/low values of the cumulative duration and the intensity/Tmin/EHF/RF. As in illustration, in Valencia (except for HWD-EHF) and Murcia (except HWD-Tmin) changes in the cumulative duration level (above/below P50) did not lead to substantial changes in the mortality risk. In contrast, different cumulative duration levels led to different RRs under the same Tmin level in cities such





**Figure 4.** For each city (*x*-axis), the relative risk (RR) is associated with each HWD (heatwave day) level. Levels are the combination of the HWD's cumulative duration and another HWD's variable (in the following, the set of these other temperature related HWD variables are referred to as Tvar), from top to bottom: HWD's recovery factor (RF), HWD's intensity, HWD's minimum temperature (Tmin), HWD's' excess heat factor (EHF). Each HWD variable is divided into two groups corresponding to values above/below the 50th percentile (P50) of the corresponding city. This subdivision results in four subgroups for each combination of cumulative duration and another Tvar. The groups above/below P50 are described as high/low values in the legend using a bivariate color scale (warm colors (red and yellow) indicate high values of Tvar while cold colors (light and dark purple) indicate low values of Tvar). Red: both Tvar and cumulative duration above P50 (high), and cumulative duration below P50 (low), light purple: Tvar below P50 (low) and cumulative duration above P50 (high), dark purple both Tvar and cumulative duration below P50 (low).

as Bilbao. In this city, increased mortality associated with HW was only observed in periods of long duration, regardless of the intensity of the event (except for RF). Badajoz is an example of a pattern consisting of an increase/decrease of the mortality risk only with a certain combination (Figure 4, fourth row, second column). In Badajoz, an increase in the mortality risk was observed only in HWDs with a simultaneous high EHF (>P50) and a large (>P50) cumulative duration meaning that the risk of mortality only increased after several days of HW conditions in which these latest HWDs had a high EHF (>P50).

# 4. Discussion

This study presents an analysis of the mortality risk associated with HW occurrence and depending on different HW characteristics in 12 province capital cities over Spain. Additionally, the effectiveness of the meteorologically widely used health indices EHF and RF to represent the HW-mortality relation has been examined. To our knowledge, this is the first assessment that analyses the mortality risk associated with different combinations of characteristics and health indices of an evenly identified set of HWs. The HWs analyzed in this study were identified in a common framework using the Spanish MetService (AEMET) definition. We identified and characterized HWs using the observational temperature gridded data set from AEMET at 5 km resolution and daily observations of mortality from 12 Spanish cities. This investigation covers a wide range of climates and population densities over Spain for the period of the extended summer months (May–September) from 1975 to 2019. We employed three distributed lag non-linear epidemiological models of increasing complexity.

Overall, our findings revealed no general patterns on how HW characteristics affect mortality in the studied cities. Due to this variation, these results may not be broadly applied to other Spanish cities, stressing the need for additional local studies.

Our study reveals an increase in the mortality risk associated with the HW occurrence in the cities with the lower HW frequencies namely Valencia, Badajoz, Bilbao, Murcia, Barcelona, and A Coruña. This increase in the mortality risk is the additive or independent effect of the HW occurrence on top of the effect associated with the presence of high temperatures. This subset of cities presents different summer temperatures and the typology of HWs affecting these cities is diverse. Except for Badajoz, these cities share a coastal location. Nevertheless, the temperatures registered on the Cantabrian coast are much lower than the Mediterranean coastal temperatures. The low HW frequency is a common feature for these cities that is not found in the other studied cities. Consequently, the HW frequency could be a factor influencing the mortality risk response to the HW occurrence. The cities with a larger HW frequency are more frequently exposed to high temperatures and may have certain adaptations, both in the population and in the city. These cities may have better-prepared infrastructures to face high temperatures. Besides, their population may have a higher awareness degree of the threat that such events pose to human health and may be more acclimatized in contrast to the cities where HWs are rare events. Another factor may be the fact that Valencia, Bilbao, Barcelona, and A Coruña also present the largest mortality rates of the cities under study.

The association between mortality risk and each HW variable did not follow a homogeneous pattern among the 12 cities examined. Each city presented a particular response to the influence of the cumulative duration combined with the intensity, minimum temperature, Excess Heat Factor or Recovery Factor on the mortality risk.

The expected performance of the health indices EHF and RF according to their definitions was an increase in the relative risk of mortality with respectively larger or lower values of the corresponding index.

This relationship was not found in all the cities. This indicates that the suitability of the EHF and the RF to assess the association between HWs and mortality in each city may be limited to cities meeting particular conditions. These conditions are probably similar to the conditions found in the cities where the RF and EHF represent the increase in mortality risk according to their definition. These cities are the Mediterranean coastal city of Valencia for the RF, and the inland city of Badajoz and Valencia for the EHF. This conclusion is particularly relevant given the recent extensive use of these indices in the scientific community to evaluate the health-related impact of HW occurrence.

As an illustration, in Valencia, the dominant characteristic in the HW-mortality relation is the increasing Tmin, and the consequent reduction of the Tmax Tmin difference, thus the reduction of the RF. In this city, the RF will show an accurate representation of the mortality risk increase given that the RF considers the response of the human body to Tmin. It is still unclear why the RF presented a different relationship with the mortality risk in other cities such as Barcelona, with similar coastal locations, climatic conditions and HW typologies. Besides, in Badajoz, where long-lasting HWs of large Tmin are associated with an increase in the mortality risk, similarly to Valencia, low values of the RF are not associated with an increase in the mortality risk.

The EHF considers the acclimatization capability of the population quantifying the intensity of sudden temperature rises. Its best performance in assessing the association between mortality and HWs was found in Valencia and Badajoz. Generally, HWs affecting Badajoz are of high Tmax and some are of simultaneously high Tmax, Tmin and intensity, contrasting with the abovementioned HW typology for Valencia. It is interesting to note that Badajoz had a similar climate and HW typology as Madrid, Sevilla, and Zaragoza where an increase in the relative risk of mortality was not observed for high EHF.

Several factors could be involved in these differences such as differences in the age distribution of the population, different population densities and mortality rates, different climates and HW typologies affecting the cities. These climatological and sociological differences may lead to a different acclimatization capability of the population. These results expose the complexity of the association between HWs, and mortality and the explanation of these differences requires detailed investigation at a local scale. A distinctive effect of the HW characteristics in different areas of the same country was also observed in Korea between rural and urban areas (Kang et al., 2020) and in Finland in Helsinki compared to its surroundings (Ruuhela et al., 2020).

Our analysis poses one step forward in analyzing the temperature-mortality relationship in Spain concerning the high-resolution temperature data set employed in conjunction with the robust modeling applied that considers different HW characteristics. Nonetheless, some limitations still are identified, such as the limited number of selected cities, or the use of all-cause mortality data instead of cause-specific mortality data, which would produce more accurate results. Moreover, in a future study, it would be advisable to consider changes in mortality risk over time, which have not been explored here, following Ordanovich et al. (2023) finding adaptability to extreme heat in the most recent decades in Spain associated with a decrease in the mortality risk attributable to extreme heat from 0.8% to 0.5% since 1988–1999.

# 5. Conclusion

The main conclusion of the study is that an analysis at the local scale is needed to correctly understand the relationship between mortality and the different HW characteristics and health-related indices RF and EHF because no general relationship has been found.

Second, in our sample, the largest mortality risks associated with the HW occurrence are observed in the cities with the lower HW frequency. This increase in the relative risk of mortality is the independent or additive effect of the HW occurrence on top of the effect of high temperatures. However, a detailed analysis is needed in each city given the fact that the factors involved in this finding may vary across the cities.

Third, the health indices widely used, the RF and the EHF, do not always adequately represent the relation between HW occurrence and mortality. This concerns the intrinsic relation found between the characteristics of the HWs and the related mortality which is different in each city. These differences could be due to several factors such as the local climatology, the characteristics of the HWs affecting the cities, the demography, the socioeconomic level, or the adaptation measures to high temperatures implemented in the city. A more profound understanding of the factors involved in the relationship between these health indices and mortality is needed to understand the reason behind their distinctive performance in cities with similar HW typologies and climates.

The results presented in this work contribute to an improved understanding of the extreme heat-health relation at a regional-to-local scale. These findings further aid in the development of improved public health strategies to protect the Spanish population from extreme heat at a local scale. Particularly, the accurate implementation of efficient heat-health warning systems adapted to the local particularities would enhance the capacity to respond to these extreme events. Despite the crucial knowledge gained in this study, a more detailed analysis of the influence of the Urban Heat Island (UHI) effect and the large thermal variability identified in the cities will be necessary to complement our understanding of the extreme heat-mortality relationship. Further investigations addressing to what extent the different factors are contributing to the relationship between HWs and mortality are needed. The city-specific particularities found in the present study reinforce the need for an assessment of the HW-mortality association in other cities of Spain but also other populated areas of the globe suffering an intensification of the HW phenomena. A recent study analyzing the local evolution of the HWs over Europe reveals that HWs affecting Germany, the Netherlands and Belgium have become more intense and present larger values of the EHF, thus a larger impact on human health, in the last decades accompanied by a large variability across the continent (Khodayar & Paredes-Fortuny, 2024) exposing the increasing threat of HWs over Europe.

# **Conflict of Interest**

The authors declare no conflicts of interest relevant to this study.

# **Data Availability Statement**

Data set ROCIO\_IBEB of daily maximum and minimum temperature was freely downloaded from AEMET (Agencia Española de Meteorología, https://www.aemet.es/en/serviciosclimaticos/cambio\_climat/datos\_dia-rios?w=2). Daily records of mortality from 12 Spanish cities were requested at INE (Instituto Nacional de Estadística).



#### Acknowledgments

The authors thank the Spanish Meteorological Service (AEMet) for the data provided for this work the data set ROCIO IBEB of daily maximum and minimum temperature. The data sets are available at https://www.aemet.es/en/ serviciosclimaticos/cambio\_climat/datos\_ diarios?w=2. The authors thank the Spanish Statistics Institute (Instituto Nacional de Estadística, INE) for providing the mortality data. This data set consisted of anonymized individual records, which were subsequently aggregated to create daily time series. The data set was collected under an official data-sharing agreement in compliance with data regulation laws and cannot be made publicly available. AMVC acknowledges funding from the Swiss National Science Foundation (TMSGI3\_211626). C. Salvador was financially supported by the ED481B-2021-122 Grant from the Xunta de Galicia, Spain. The EPhysLab group was cofounded by Xunta de Galicia, Consellería de Cultura, Educación e Universidade, under project ED431C 2021/44 "Programa de Consolidación e Estructuración de Unidades de Investigación Competitivas" and the European Union ERDF a way of making Europe. C. Salvador thanks the computing resources provided by CESGA (Centro de Supercomputación de Galicia). C. Salvador and A.M. Vicedo-Cabrera acknowledges TED2021-129152B-C43 project by the Ministerio de Ciencia, Innovación y Universidades (Spain) and the European Union "ERDF A way of making Europe" "NextGenerationEU"/ PRTR. C.Salvador and A.M. Vicedo-Cabrera are part from Ibero-American Women Network for Climate Action (IBWoClimA). The Mediterranean Centre for Environmental Studies (CEAM) is partly supported by Generalitat Valenciana. The contribution of Laura Paredes-Fortuny and Samira Khodayar Pardo was supported by the program Generació Talent of Generalitat Valenciana (CIDEGENT/2018/017).

#### References

- Achebak, H., Garcia-Aymerich, J., Rey, G., Chen, Z., Méndez-Turrubiates, R. F., & Ballester, J. (2023). Ambient temperature and seasonal variation in inpatient mortality from respiratory diseases: A retrospective observational study. *The Lancet Regional Health—Europe*, 35, 100757. https://doi.org/10.1016/j.lanepe.2023.100757
- AEMET Área de Climatología y Aplicaciones Operativas. Olas de calor en España des de 1975. (2023). Retrieved from https://www.aemet.es/ documentos/es/conocermas/recursos\_en\_linea/publicaciones\_y\_estudios/estudios/Olas\_calor/olas\_calor\_actualizacionoct2023.pdf
- Amblar-Francés, M. P., Ramos-Calzado, P., Sanchis-Lladó, J., Hernanz-Lázaro, A., Peral-García, M. C., Navascués, B., et al. (2020). High resolution climate change projections for the Pyrenees region [Dataset]. Advances in Science and Research, 17, 191–208. https://doi.org/10. 5194/asr-17-191-2020
- Arsad, F. S., Hod, R., Ahmad, N., Ismail, R., Mohamed, N., Baharom, M., et al. (2022). The impact of heatwaves on mortality and morbidity and the associated vulnerability factors: A systematic review. *International Journal of Environmental Research and Public Health*, 19(23), 16356. https://doi.org/10.3390/ijerph192316356
- Ballester, J., Quijal-Zamorano, M., Turrubiates, R. F., Pegenaute, F., Herrmann, F. R., Robine, J. M., et al. (2023). Heat-related mortality in Europe during the summer of 2022. *Nature Medicine*, 29(7), 1857–1866. https://doi.org/10.1038/s41591-023-02419-z
- Barriopedro, D., García-Herrera, R., Ordóñez, C., Miralles, D. G., & Salcedo-Sanz, S. (2023). Heat waves: Physical understanding and scientific challenges. *Reviews of Geophysics*, 61(2), e2022RG000780. https://doi.org/10.1029/2022rg000780
- Carmona, R., Linares, C., Ortiz, C., Mirón, I. J., Luna, M. Y., & Díaz, J. (2017). Spatial variability in threshold temperatures of heat wave mortality: Impact assessment on prevention plans. *International Journal of Environmental Health Research*, 27(6), 463–475. https://doi.org/10. 1080/09603123.2017.1379056
- D'Ippoliti, D., Michelozzi, P., Marino, C., de'Donato, F., Menne, B., Katsouyanni, K., et al. (2010). The impact of heat waves on mortality in 9 European cities: Results from the EuroHEAT project. *Environmental Health*, 9(1), 37. https://doi.org/10.1186/1476-069X-9-37
- Díaz, J., Carmona, R., Mirón, I. J., Luna, M. Y., & Linares, C. (2018). Time trend in the impact of heat waves on daily mortality in Spain for a period of over thirty years (1983–2013). *Environment International*, *116*, 10–17. https://doi.org/10.1016/j.envint.2018.04.001
- Gasparrini, A. (2011). Distributed lag linear and non-linear models in R: The package dlnm. JSS Journal of Statistical Software, 43(8). https://doi.org/10.18637/jss.v043.i08
  Gasparrini, A., Armstrong, B., & Kenward, M. G. (2010). Distributed lag non-linear models. Statistics in Medicine, 29(21), 2224–2234. https://
- doi.org/10.1002/sim.3940
- Guo, Y., Gasparrini, A., Armstrong, B. G., Tawatsupa, B., Tobias, A., Lavigne, E., et al. (2017). Heat wave and mortality: A mortality, multicountry study. *Environmental Health Perspectives*, 125, 8. https://doi.org/10.1289/EHP1026
- Huang, J., Wang, J., & Yu, W. (2014). The lag effects and vulnerabilities of temperature effects on cardiovascular disease mortality in a subtropical climate zone in China. International Journal of Environmental Research and Public Health, 11, 4–3994. https://doi.org/10.3390/ ijerph110403982
- IPCC, Seneviratne, S. I., Zhang, X., Adnan, M., Badi, W., Dereczynski, C., et al. (2021). Weather and climate extreme events in a changing climate. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. P'ean, et al. (Eds.), Climate change 2021: The physical science basis. Contribution of working group I to the sixth assessment report of the intergovernmental panel on climate change (pp. 1513–1766). Cambridge University Press. https://doi.org/10.1017/9781009157896.013
- Kang, C., Park, C., Lee, W., Pehlivan, N., Choi, M., Jang, J., & Kim, H. (2020). Heatwave-related mortality risk and the risk-based definition of heat wave in South Korea: A nationwide time-series study for 2011–2017. *International Journal of Environmental Research and Public Health*, 17(16), 1–12. https://doi.org/10.3390/ijerph17165720
- Karl, T. R., & Knight, R. W. (1997). The 1995 Chicago heat wave: How likely is a recurrence? Bulletin of the American Meteorological Society, 78(6), 1107–1119. https://doi.org/10.1175/1520-0477(1997)078<1107:TCHWHL>2.0.CO;2
- Khodayar, S., & Paredes-Fortuny, L. (2024). Uneven evolution of regional European summer heatwaves under climate change. Weather and Climate Extremes, 43, 100648. https://doi.org/10.1016/j.wace.2024.100648
- Kollanus, V., Tiittanen, P., & Lanki, T. (2021). Mortality risk related to heatwaves in Finland—Factors affecting vulnerability. Environmental Research, 201, 111503. https://doi.org/10.1016/j.envres.2021.111503
- Linares, C., Sánchez, R., Mirón, I. J., & Díaz, J. (2015). Has there been a decrease in mortality due to heat waves in Spain? Findings from a multicity case study. *Journal of Integrative Environmental Sciences*, 12(2), 153–163. https://doi.org/10.1080/1943815X.2015.1062032
- Lionello, P., & Scarascia, L. (2018). The relation between climate change in the Mediterranean region and global warming. *Regional Environmental Change*, 18(5), 1481–1493. https://doi.org/10.1007/s10113-018-1290-1
- Lo, Y. T. E., Mitchell, D. M., Thompson, R., O'Connell, E., & Gasparrini, A. (2022). Estimating heat-related mortality in near real time for national heatwave plans. *Environmental Research Letters*, 17(2), 024017. https://doi.org/10.1088/1748-9326/ac4cf4
- López-Bueno, J. A., Díaz, J., Sánchez-Guevara, C., Sánchez-Martínez, G., Franco, M., Gullón, P., et al. (2020). The impact of heat waves on daily mortality in districts in Madrid: The effect of sociodemographic factors. *Environmental Research*, 190, 109993. https://doi.org/10.1016/j. envres.2020.109993
- Lorenzo, N., Díaz-Poso, A., & Royé, D. (2021). Heatwave intensity on the Iberian Peninsula: Future climate projections. *Atmospheric Research*, 258, 105655. https://doi.org/10.1016/j.atmosres.2021.105655
- McKenna, Z. J., Foster, J., Atkins, W. C., Belval, L. N., Watso, J. C., Jarrard, C. P., et al. (2023). Age alters the thermoregulatory responses to extreme heat exposure with accompanying activities of daily living. *Journal of Applied Physiology*, 135(2), 445–455. https://doi.org/10.1152/ japplphysiol.00285.2023
- MedECC. (2020). In W. Cramer, J. Guiot, & K. Marini. (Eds.), Climate and environmental change in the Mediterranean Basin—Current situation and risks for the future. First Mediterranean assessment report, union for the Mediterranean, (p. 632). UNEP/MAP, https://doi.org/10.5281/ zenodo.7224821
- Miron, I. J., Linares, C., Montero, J. C., Criado-Alvarez, J. J., & Díaz, J. (2015). Changes in cause-specific mortality during heat waves in central Spain, 1975–2008. International Journal of Biometeorology, 59(9), 1213–1222. https://doi.org/10.1007/s00484-014-0933-2
- Moraes, S. L., de Almendra, R., & Barrozo, L. V. (2022). Impact of heat waves and cold spells on cause-specific mortality in the city of São Paulo, Brazil. International Journal of Hygiene and Environmental Health, 239, 113861. https://doi.org/10.1016/j.ijheh.2021.113861
- Nairn, J., & Fawcett, R. (2014). The excess heat factor: A metric for heatwave intensity and its use in classifying heatwave severity. International Journal of Environmental Research and Public Health, 12(1), 227–253. https://doi.org/10.3390/ijerph120100227
- Ordanovich, D., Tobías, A., & Ramiro, D. (2023). Temporal variation of the temperature-mortality association in Spain: A nationwide analysis. *Environmental Health*, 22(1), 5. https://doi.org/10.1186/s12940-022-00957-6



- Paredes-Fortuny, L., & Khodayar, S. (2023). Understanding the magnification of heatwaves over Spain: Relevant changes in the most extreme events. *Weather and Climate Extremes*, 42, 100631. https://doi.org/10.1016/j.wace.2023.100631
- Pereira, S. C., Marta-Almeida, M., Carvalho, A. C., & Rocha, A. (2017). Heat wave and cold spell changes in Iberia for a future climate scenario. *International Journal of Climatology*, 37(15), 5192–5205. https://doi.org/10.1002/joc.5158
- Robine, J.-M., Cheung, S. L., Roy, S. L., Oyen, H. V., Griffiths, C., Michel, J.-P., & Herrmann, F. R. (2008). Death toll exceeded 70,000 in Europe during the summer of 2003. *Comptes Rendus Biologies*, 331(2), 171–178. https://doi.org/10.1016/j.crvi.2007.12.001
- Roldán, E., Gómez, M., Pino, M. R., Pórtoles, J., Linares, C., & Díaz, J. (2016). The effect of climate-change-related heat waves on mortality in Spain: Uncertainties in health on a local scale. *Stochastic Environmental Research and Risk Assessment*, 30(3), 831–839. https://doi.org/10. 1007/s00477-015-1068-7
- Royé, D., Codesido, R., Tobías, A., & Taracido, M. (2020). Heat wave intensity and daily mortality in four of the largest cities of Spain. *Environmental Research*, 182, 109027. https://doi.org/10.1016/j.envres.2019.109027
- Ruiz-Páez, R., Díaz, J., López-Bueno, J. A., Navas, M. A., Mirón, I. J., Martínez, G. S., et al. (2023). Does the meteorological origin of heat waves influence their impact on health? A 6-year morbidity and mortality study in Madrid (Spain). Science of the Total Environment, 855, 158900. https://doi.org/10.1016/j.scitotenv.2022.158900
- Ruuhela, R., Votsis, A., Kukkonen, J., Jylhä, K., Kankaanpää, S., & Perrels, A. (2020). Temperature-related mortality in Helsinki compared to its surrounding region over two decades, with special emphasis on intensive heatwaves. *Atmosphere*, 12(1), 46. https://doi.org/10.3390/ atmos12010046
- Silveira, I. H., Cortes, T. R., Bell, M. L., & Junger, W. L. (2023). Effects of heat waves on cardiovascular and respiratory mortality in Rio de Janeiro, Brazil. *PLoS One*, *18*(3), e0283899. https://doi.org/10.1371/journal.pone.0283899
- Tobías, A., Royé, D., & Iñiguez, C. (2023). Heat-attributable mortality in the summer of 2022 in Spain. *Epidemiology*, 34(2), e5–e6. https://doi.org/10.1097/EDE.000000000001583
- Viceto, C., Pereira, S. C., & Rocha, A. (2019). Climate change projections of extreme temperatures for the Iberian Peninsula. *Atmosphere*, 10(5), 229. https://doi.org/10.3390/atmos10050229
- WMO-No 1203. (2017). WMO guidelines on the calculation of climate normals. In *Purpose*. Chairperson, Publications Board World Meteorological Organization (WMO).
- Yang, J., Yin, P., Sun, J., Wang, B., Zhou, M., Li, M., et al. (2019). Heatwave and mortality in 31 major Chinese cities: Definition, vulnerability and implications. *Science of the Total Environment*, 649, 695–702. https://doi.org/10.1016/j.scitotenv.2018.08.332
- Zhao, Q., Guo, Y., Ye, T., Gasparrini, A., Tong, S., Overcenco, A., et al. (2021). Global, regional, and national burden of mortality associated with non-optimal ambient temperatures from 2000 to 2019: A three-stage modelling study. *The Lancet Planetary Health*, 5(7), e415–e425. https:// doi.org/10.1016/S2542-5196(21)00081-4